Author's response to reviews

Title: The short- and long-term outcomes of radical antegrade modular pancreatosplenectomy for adenocarcinoma of the body and tail of the pancreas

Authors:

Masaaki Murakawa (m-murakawa@kcch.jp)
Toru Aoyama (aoyamat@kcch.jp)
Masahiro Asari (asari@kcch.jp)
Yusuke Katayama (ykatayama@kcch.jp)
Koichiro Yamaoku (k-yamaoku@kcch.jp)
Amane Kanazawa (amanex2009@gmail.com)
Akio Higuchi (a-higuchi@kcch.jp)
Manabu Shiozawa (shiozawam@kcch.jp)
Satoshi Kobayashi (kobayashis@kcch.jp)
Makoto Ueno (uenom@kcch.jp)
Manabu Morimoto (m-morimoto@kcch.jp)
Naoto Yamamoto (naoto_y@yokohama-cu.ac.jp)
Takaki Yoshikawa (yoshikawat@kcch.jp)
Yasushi Rino (rino@yokohama-cu.ac.jp)
Munetaka Masuda (mmasuda@yokohama-cu.ac.jp)
Soichiro Morinaga (morinagas@kcch.jp)

Version: 2 Date: 21Oct 2015
Author’s response to reviews:

Editor-in-Chief
BMC surgery

Dear editor,

We are grateful to you for kindly reviewing our manuscript. According to your instructive comments, we have revised the manuscript with some revision. All corrections were marked by yellow and underline in the text. The followings are responses to comments. Please consider this manuscript again for publication in the BMC Surgery.

Thank you again for your kind consideration, and we are looking forward to your reply.

Respectfully yours,

Toru Aoyama (Corresponding author)
Department of Gastrointestinal Surgery, Kanagawa Cancer Center
2-3-2 Nakao, Asahi-Ku, Yokohama 241-8515, Japan
Tel:+81-45-520-2222
Fax:+81-45-520-2226
E-mail: aoyamat@kcch.jp

Comment1:

Please kindly include all authors' emails on title page.

Answer to comment 1:

As pointed out, we added all author’s email address on title page. (Page 1, Line 16 to Page 2, Line 30)

Comment2:

Research involving human subjects (including human material or human data) that is reported in the manuscript must have been performed with the approval of an appropriate ethics committee. Research carried out on humans must be in compliance with the Helsinki Declaration (http://www.wma.net/en/30publications/10policies/b3/index.html). A statement to this effect must appear in the Methods section of the manuscript, including the name of the body which gave approval, with a reference number where appropriate.

Answer to comment 2:

Thank you for the important comments. As pointed out, we added the sentence about the approval of institutional review board of our center. Moreover, this study was in compliance with the Declaration of Helsinki (Page 9, Line 169 to Page 9, Line 171).